

# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2020GL090068

#### **Key Points:**

- We model potential evasion of terrestrially sourced headwater CO<sub>2</sub> from over 98,000 river and lake units in the Connecticut River watershed
- Lakes are responsible for approximately 25%–30% of potential CO<sub>2</sub> evasion, as influenced by streamflow and stream order
- Lake CO<sub>2</sub> evasion efficiency is a function of residence time and size, where larger lakes evade functionally 100% of CO<sub>2</sub> from upstream

#### **Supporting Information:**

Supporting Information S1

#### Correspondence to:

C. Brinkerhoff, cbrinkerhoff@umass.edu

#### **Citation**:

Brinkerhoff, C. B., Raymond, P. A., Maavara, T., Ishitsuka, Y., Aho, K. S., & Gleason, C. J. (2021). Lake morphometry and river network controls on evasion of terrestrially sourced headwater CO<sub>2</sub>. *Geophysical Research Letters*, *48*, e2020GL090068. https://doi.org/10.1029/2020GL090068

Received 29 JUL 2020 Accepted 19 NOV 2020

# Lake Morphometry and River Network Controls on Evasion of Terrestrially Sourced Headwater CO<sub>2</sub>

C. B. Brinkerhoff<sup>1</sup>, P. A. Raymond<sup>2</sup>, T. Maavara<sup>2</sup>, Y. Ishitsuka<sup>1</sup>, K.S. Aho<sup>2</sup>, and C. J. Gleason<sup>1</sup>

<sup>1</sup>Department of Civil and Environmental Engineering, University of Massachusetts, Amherst, MA, USA, <sup>2</sup>School of the Environment, Yale University, New Haven, CT, USA

**Abstract** Lakes are central components of the inland water system distinct from, yet inextricably connected to, river networks. Currently, existing network-scale biogeochemistry research, although robust, typically treats each of these components separately or reductively. Here, we incorporate lake morphometry into a fully connected stream/lake network for the Connecticut River watershed and model potential evasion of terrestrially sourced headwater CO<sub>2</sub> as transported through the network, ignoring in-stream production. We found that approximately 25%–30% of total potential soil CO<sub>2</sub> evasion occurs in lakes, and percent evasion is inversely related to streamflow. A lake's ability to evade CO<sub>2</sub> is controlled by residence time and size: most lakes with residence time over 7 days or surface area greater than 0.004 km<sup>2</sup> evade functionally all terrestrial CO<sub>2</sub> entering from upstream, precluding further downstream transport. We conclude that lakes are important for soil CO<sub>2</sub> degassing and that this coupled river/lake approach is promising for CO<sub>2</sub> studies henceforth.

**Plain Language Summary** River networks are both delivery systems and active transformers of constituents (sediment, nutrients, biota, heat) as they carry them to the sea. An important and overlooked component of these networks are lakes: lakes dominate total water storage of surface water in river networks and have vastly different hydraulics than rivers. Despite this knowledge, most greenhouse gas (GHG) research addresses lakes and rivers separately or reductively. To address this, we built a fully connected model for over 98,000 rivers and lakes in the Connecticut River watershed and found that lakes emit substantially more carbon dioxide than rivers per unit (potentially 25%–30% of the total emission from the system), and that almost all lakes emit almost all the carbon dioxide that enters their waters. These findings demonstrate the importance of connected lakes to drainage network GHG exchange and corroborate the need to better integrate lakes into our understandings of GHG emissions from freshwater systems in a formalized framework.

# 1. Introduction

Inland waters play an important role in the global carbon cycle, contributing significant greenhouse gas (GHG) emissions to the atmosphere (Bastviken et al., 2011; Cole et al., 2007; Raymond et al., 2013). River networks are generally conceptualized as active biogeochemical reactors that mix, store, and evade GHGs and constituents transported from upstream together with those generated from in-stream production (Cole et al., 2007; see also; Raymond et al., 2016; Zarnetske et al., 2018). Ultimately, inland waters store, evade, and transport over half of the carbon that they receive from the terrestrial ecosystem before reaching the oceans (Hotchkiss et al., 2015), thereby playing a fundamental role in global carbon processes.

Running inland waters (hereafter termed 'rivers') are generally supersaturated with GHGs and exhibit a net evasive flux of these gases from water to air (Cole & Caraco, 2001; Jones et al., 2003). This flux  $[M/L^2T]$  is relatively easy to calculate with in situ knowledge of the gas concentration gradient between the water  $[gas]_{water}$  and the air  $[gas]_{air}$   $[M/L^3]$  and the gas transfer velocity k [L/T] (Equation 1).

$$Flux = k \left( \left[ gas \right]_{water} - \left[ gas \right]_{kir} \right)$$
(1)

Because k is largely a function of surface water turbulence, and thus river channel hydraulics, landscape geomorphology (which drives channel hydraulics) fundamentally regulates how k manifests across streams

© 2020. American Geophysical Union. All Rights Reserved. and rivers (Raymond et al., 2012; Ulseth et al., 2019). More broadly, geomorphology and hydraulics affect GHG evasion from inland waters in two ways: first, while gas evasion from rivers is driven by surface turbulence (Zappa et al., 2007), evasion from lakes and reservoirs (hereafter referred to as 'lakes') is additionally influenced by turbulence from convection (particularly in small ponds - Holgerson et al., 2016; Read et al., 2012). The much longer residence times of lakes also allows for more gas evasion to occur (Catalán et al., 2016; Cheng & Basu, 2017). Second, in-stream GHGs are sourced from multiple landscape components. As a representative GHG, carbon dioxide (CO<sub>2</sub>) is terrestrially sourced from soil respiration or decomposition in groundwater (Duvert et al., 2018; Hotchkiss et al., 2015). Terrestrial CO<sub>2</sub>'s influence on stream evasion decreases downstream as the terrestrial edge to water volume ratio decreases and in-stream metabolic and abiotic processes become relatively more important due to longer residence times in larger rivers (Battin et al., 2008; Hotchkiss et al., 2015; Marx et al., 2017; Öquist et al., 2009; Raymond et al., 2016). Further, the majority of terrestrial  $CO_2$  in the headwaters is thought to come from soil respiration (e.g. Hope et al., 2004; Johnson et al., 2008). Taken in aggregate, in-stream  $CO_2$  is a complex manifestation of terrestrial inputs, in-stream processes, evasion to the atmosphere, and transport mechanics that are all in some way functions of the local landscape and discharge (Liu & Raymond, 2018). Thus, the heterogenous nature of a watershed's geomorphology makes it difficult to scale GHG evasion measurements to entire river networks.

In spite of these difficulties, researchers have attempted to upscale in situ GHG evasion measurements to either river networks (Borges et al., 2015; Butman & Raymond, 2011; Horgby et al., 2019; Hu et al., 2016; Lauerwald et al., 2015; Raymond et al., 2013) or large waterbodies data sets composed of thousands of lakes, reservoirs, and/or wetlands (Deemer et al., 2016; DelSontro et al., 2018; Holgerson & Raymond, 2016; Lauerwald et al., 2019; Soued et al., 2016). We have already established that different physical processes control gas evasion in rivers and lakes. However, most work on gas evasion at network scales neglects to acknowledge the two systems' intrinsic connectivity via transport mechanics and network topology (Crawford et al., 2014; Fergus et al., 2017; Gardner et al., 2019; Wetzel, 2001). This disconnect is beginning to be addressed in related subfields like sediment transport (Czuba & Foufoula-Georgiou, 2015; Czuba et al., 2017) and nutrient transport (Bertuzzo et al., 2017; Schmadel et al., 2018, 2019; Wollheim et al., 2008). A preliminary treatment of nitrous oxide emissions from rivers versus reservoirs has also been performed (Maavara et al., 2019). However, without a spatially explicit treatment of river/lake connectivity in the context of potential GHG evasion, our ability to accurately constrain network-scale evasion is limited.

We sought to answer the following questions, using terrestrially derived  $CO_2$  sourced from headwaters as a representative GHG and a model of a fully coupled river/lake network: (1) How do lakes affect the evasion of terrestrially derived  $CO_2$ , and how does this compare to evasion of this same  $CO_2$  from rivers? and (2) how do lake residence times and sizes influence their ability to evade  $CO_2$  transported from upstream (termed 'CO<sub>2</sub> evasion efficiency')? Using the Connecticut River's watershed, we explicitly incorporate lake morphometry into a network-scale potential evasion model with full conservation of mass and momentum and across a full hydrological flow regime to answer these questions (after validating the model – Section 3.1).

# 2. Methods

To construct the model outlined above, we merged hydrography with lake morphometry, streamflow routing, and a gas advection/evasion model. First, we outline our data set (Section 2.1), then we describe our  $CO_2$  potential evasion model (Section 2.2), and finally we describe the experiment (Section 2.3).

## 2.1. Hydrography

We used the United States Geological Survey (USGS)'s NHDPlus High Resolution hydrography (NHD HR) to represent river network topology in the Connecticut River watershed. The Connecticut River is the largest river in the northeastern United States, draining nearly 30,000 km<sup>2</sup> from the Quebec border to its outlet in Long Island Sound. This data set includes waterbodies, a river network, and artificial 'throughflow' lines that topologically connect the waterbodies to the river network. We spatially mapped the attributes assigned to each waterbody classified by the USGS as a 'lake/reservoir' (treated as synonymous in the NHD HR and this study) to the artificial throughflow lines, such that the unique characteristics of each lake were embedded in the river network. It is worth stressing that reservoirs can have distinct morphometry from natural





**Figure 1.** Conceptual representation of the model simulating potential evasion of terrestrial CO<sub>2</sub> (right) from our hydrography for the Connecticut River (left – U.S. Geological Survey, 2019b). River thickness corresponds to stream order, of which only third-eighth orders are mapped for visualization. Note that lateral CO<sub>2</sub> inputs to all rivers/lakes larger than first order are ignored.

lakes, but we use the term 'lake' interchangeably for easier readability. We make no claims in this study about differential impacts from natural versus anthropogenic waterbodies. Ultimately, our river network contained 98,254 reaches including throughflow lines that represent 14,884 total lakes. However, we still lack information on two crucial variables necessary for the  $CO_2$  advection/evasion model (Section 2.2): lake volume and river/lake discharge.

The USGS provides estimates for lake volume in their lower resolution hydrography (NHD), which they modeled using surrounding topography (Hollister & Milstead, 2010). We ran their same lake volume generation algorithm on our higher resolution data set's lakes that were  $> 0.1 \text{ km}^2$  in surface area. For those smaller than 0.1 km<sup>2</sup>, we used a statistical scaling function to predict volume from surface area (Cael et al., 2017, Text S1). Many larger lakes have multiple intersecting throughflow lines in the NHD HR, effectively segmenting the lake. When assigning lake volumes in these scenarios, we mapped relative portions of the lake to each throughflow line (Figure 1, Text S1).

For river/lake discharge, mean annual flow is not sufficient for our experiment: we are interested in a full flow regime reflecting not only average streamflow but also extreme floods, drought events, and the flashiness of the streamflow regime. Therefore, we built flow duration curves (which constitute the probability of exceedance of a range of streamflows) for every reach in the network. To do so, we downscaled runoff forcing data and routed it through our river network via the Hillslope River Routing model (HRR- Beighley et al., 2009) to generate daily discharge estimates at every reach in the network. We used daily runoff estimates from the 'Global Reach-Level A Priori Discharge Estimates for SWOT' data set (GRADES – Lin et al., 2019) for coarse-scale forcing data and used 1979–1989 as a case study. This process is outlined in Text S2. We validated our modeled flow duration curves at 87 stream gauges in the watershed for this time period (Figure S1, data accessed via the USGS's dataRetrieval R package), finding near perfect recovery of discharges > 1 m<sup>3</sup>/s and strong recovery of flow in the smallest streams (<1 m<sup>3</sup>/s, Figure S1).

#### 2.2. CO<sub>2</sub> Advection/Evasion Model

Because our focus is on the role of lake/river connectivity, we are specifically interested in tracking the fate and transport of headwater  $CO_2$  and we ignore downstream terrestrial inputs and in-stream processes,

acknowledging that a full picture of carbon cycling should include these processes. We made this choice because we could not defend either where or how much terrestrial  $CO_2$  to add further downstream. We can defend adding terrestrial  $CO_2$  to the headwaters (Hotchkiss et al., 2015; Liu & Raymond, 2018; Marx et al., 2017; Winterdahl et al., 2016). Thus, we leave differential downstream  $CO_2$  inputs and in-stream  $CO_2$  generation for future work.

To model potential evasion of terrestrial  $CO_2$  from the river network, we created a model that tracks distinct 'parcels' of  $CO_2$  (i.e. the Lagrangian specification of the flow field) as they simultaneously evade gas to the atmosphere, move downstream, and interact with other  $CO_2$  parcels from intersecting streams (Figure 1). This approach is in line with advection/evasion modeling previously performed for the Amazon River (Abril et al., 2014).

Conceptually, we introduce some amount of  $CO_2$  into the headwaters of the network (the term 'headwater' is used interchangeably with first order here and throughout). Next,  $CO_2$  is evaded within each reach via Equation 2 (Text S3 for derivation), where *i* refers to a specific reach,  $[CO_2]_{i-1}$  is the inflowing  $CO_2$  concentration from upstream  $[M/L^3]$ , and HRT is the hydraulic residence time [T]. Note  $[CO_2]_{i-1}$  is sourced laterally in the first order. These terms are all defined explicitly for every reach in the network but *k* and HRT were calculated differently for lakes and rivers to reflect the distinct hydrological processes occurring in the two environments (Text S4). In short, stream and river *k* is modeled as a function of water column turbulence (Raymond et al., 2012) while lake *k*, which is largely wind and convection driven, is modeled using surface area as a reasonable proxy for wind shear and convection in the surface mixed layer (Raymond et al., 2013; Read et al., 2012):

$$\left[\operatorname{CO}_{2}\right]_{\operatorname{evaded},i} = \left[\operatorname{CO}_{2}\right]_{i-1} - \left\{\left[\operatorname{CO}_{2}\right]_{i-1} e^{-k_{i} * \operatorname{HRT}_{i}}\right\}$$
(2)

Any remaining in-stream  $CO_2$  is then transported to the next reach immediately downstream. This process is continued for all downstream reaches until the most downstream river in the network is reached. When there are multiple inflows to a reach, the  $CO_2$  inflow is 'diluted' and calculated as the mean  $CO_2$  concentration across the inflows, weighted by discharge to favor larger rivers (Figure 1).

While we used an arbitrary  $CO_2$  concentration to report ultimate results of the role of lakes, we validated our model by forcing it with field measurements of in-stream  $CO_2$  concentrations made at three sub-catchments that met our validation requirements (Text S5, Figure S2): (1) the Salmon River (Aho & Raymond, 2019), (2) the West Branch Swift River, and (3) the Harvard Forest (both using USGS measurements again accessed via the dataRetrieval R package). To validate, we set the model input  $CO_2$  concentration as measured in the field at the corresponding (and spatially explicit) stream reach and ran our model at the flow corresponding to the flow at the time of the field measurement. We then compared downstream model outputs to separate field validation measurements that were made at downstream locations later in time, thus allowing a validation of  $CO_2$  passing downstream along a river network.

#### 2.3. Experimental Design

With a validated model in hand, we introduced an arbitrary parcel of  $CO_2$  (4,987 µatm or 10 mg/L at a uniform stream temperature of 15 °C) into every first-order stream to represent  $CO_2$  terrestrially sourced from soil respiration in the headwaters. We ignore terrestrial  $CO_2$  inputs from groundwater for the sake of model parsimony, acknowledging that the headwaters are highly sensitive to groundwater inputs (Duvert et al., 2018). We also ignore the effects of the carbonate buffering system, which is well-known to influence riverine  $CO_2$  emissions (Stets et al., 2017).

We then ran the model for a suite of 'characteristic discharges' extracted from the flow duration curves at all 98,254 reaches in the network. 'Characteristic discharges' were defined using exceedance probabilities from each reach's unique flow duration curve (i.e.  $Q_{10}$  is the discharge exceeded 10% of the time in a reach), and include  $Q_2$ ,  $Q_5$ ,  $Q_{10}$ ,  $Q_{15}$ ,  $Q_{25}$ ,  $Q_{35}$ ,  $Q_{50}$ ,  $Q_{65}$ ,  $Q_{75}$ ,  $Q_{90}$ ,  $Q_{95}$ ,  $Q_{98}$ , and  $Q_{\text{mean}}$ . Total network evasion was quantified as the sum of all CO<sub>2</sub> evaded into the atmosphere across all reaches. The percentage of evasion occurring in lakes was calculated as the ratio of total evasion from lakes to total network evasion. Finally, we







**Figure 2.** Modeled versus observed  $CO_2$  concentrations for (a) all validation measurements and (b) after filtering for high flows, here defined as < 30% exceedance probability.

defined ' $CO_2$  evasion efficiency' as the sum of  $CO_2$  evaded from all throughflow lines within a lake divided by the sum of  $CO_2$  evaded and transported downstream.

# 3. Results and Discussion

#### 3.1. Advection/Evasion Model Predicts Downstream Patterns in Terrestrial CO<sub>2</sub>

Field measurements of downstream  $CO_2$  concentrations are systematically underestimated by our model at the four sites tested when considering all flows (Figure 2a). However, when we filter for only high flow events (defined here as flow that is less than an exceedance probability of 30%), the model successfully reproduces observed in-stream  $CO_2$  concentrations (Figure 2b). Below, we explore why this is the case and why we assert that Figure 2b successfully validates our model.

Raymond et al. (2016) and Zarnetske et al. (2018) suggested that river systems likely swing back and forth between transport and reactor states depending upon current flow conditions and water temperature. During low flow events, high residence times allow in-stream processes to occur and solute concentration is fundamentally influenced by biological removal/production. However, during flood events, residence times are short enough that there is not enough time for stream ecosystems to process solutes before advection, merely 'shunting' them downstream (Raymond et al., 2016). Similar relationships between water residence time and evasion have been studied in the specific context of  $CO_2$  evasion (Duvert et al., 2018; Maavara et al., 2020; Marx et al, 2017). Therefore, because we modeled only advection/evasion of terrestrial  $CO_2$ sourced from the headwaters and ignored in-stream processes, a fair validation must compare the model outputs to field measurements with as little influence from in-stream process as possible (i.e. high flows as previously established). Under this scenario, Figure 2b successfully validates our model. Note that this does not inhibit our ability to run the model on the entire flow regime, as it will simply produce evasion of terrestrially sourced  $CO_2$  and not total in-stream evasion at lower flows, but our model cannot explicitly disentangle these sources.

#### 3.2. Lakes Play an Outsized Role in Potential Evasion of Terrestrial CO<sub>2</sub>

At mean annual flow, about 28% of potential evasion of terrestrial  $CO_2$  that enters the network at first-order terrestrial-aquatic interfaces occurs in lakes (green line in Figure 3a, Table S1). This occurs despite rivers outnumbering lakes by a factor of nearly 5.8 (at mean annual flow): the far longer residence times of lakes allow for significantly more  $CO_2$  to be evaded to the atmosphere than in rivers. There is a non-linear





**Figure 3.** Percent of the total potential evasion of terrestrial  $CO_2$  occurring in lakes in (a) the entire Connecticut River network at mean annual flow (green line) and across a full flow regime (orange points) and (b) stratified by order. (c) The relative importance of lakes within each stream order, normalized by the wetted surface area of that order, for the full flow regime and mean annual flow (green lines). Note that 50th percentile flows correspond to median, not mean, flow. (b) Suggests that the first-order trend dominates the network-wide trend, while orders 2–5 all exhibit similar trends. We suggest that order 6's trend is an artifact of its small number of lakes (16 in total, Figure S3).

relationship between discharge and the relative importance of lakes to potential evasion in the system (Figure 3a, orange points): lake influence increases marginally to 29% at flood flows but decreases slightly to approximately 25% at low flows. While rivers still dominate the total evasion budget at over 70% of total evasion, the unit evasion of lakes is much greater than streams.

The position of lakes in the network is also important. We find that lake importance fundamentally varies by stream order and streamflow, with more relative lake efflux in the first order at high flows, and more relative efflux in other orders at low-to-mid flows (Figure 3b). Across the entire flow regime, 'unit lake influence' (normalized by surface area) monotonically increases with stream order (Figure 3c), suggesting that the bigger lakes downstream (Figure S3) are more efficient evaders of terrestrial  $CO_2$  (Figure 3b and Table S2). This increasing lake importance downstream occurs despite most absolute evasion occurring in the first order (Figure 4c). There are no lakes in the system's seventh or eighth orders.

Our results show that high flows expand the wetted extent of the network and substantially increase the surface area of the first order (e.g.  $Q_{10}$  has 181 km<sup>2</sup> of first-order rivers and lakes, while  $Q_{90}$  has only 156 km<sup>2</sup>).



# **Geophysical Research Letters**



**Figure 4.**  $CO_2$  evasion efficiency versus lake HRT (a) and size (b) for each lake at mean annual flow, binned by quantiles for visualization's sake. (c) Total evasion by order (at mean annual flow) and normalized by order surface area. There are no lakes in the seventh or eighth orders, though there is still almost 1 mg/L × km<sup>2</sup> of evasion occurring.

This likely causes the observed increases in the relative importance of first-order lakes at high flows (Figure 3b). Because of the sheer number of elements in the first order (both rivers and lakes), we suspect that the first order is largely responsible for the network-wide trend whereby lake importance increases with flow (i.e. Figure 3a orange points mimic the first-order green points in Figure 3b, but do not mimic any of the other stream orders). We conclude this because in all other stream orders (except the sixth), relative lake importance decreases with flow. The sixth-order trend is likely an artifact of its small number of lakes (16 in total, Figure S3). Thus we conclude that at high flows  $CO_2$  parcels move downstream through the system quickly and diminish the effect of the long residence times in lakes. Finally, unit lake influence is largely constant within each order, regardless of flow across orders (Figure 3c).

# 3.3. Lake Morphometry and CO<sub>2</sub> Evasion Efficiency

Our results show that lake morphometry controls total evasion of terrestrial  $CO_2$  from individual lakes based on their residence time and size. We calculated  $CO_2$  evasion efficiency for every lake at mean annual flow and then compared this metric against lake surface area (Figure 4a) and mean annual HRT (Figure 4b), binning the results by quantiles to visualize similarly sized evasion efficiency distributions.  $CO_2$  evasion efficiency generally scales with HRT (Figure 4a), which is intuitive and follows the structure of the advection/evasion model (Equation 2). There are effectively no lakes that are simultaneously low HRT and high efficiency, and likewise there are no high HRT, low efficiency lakes. Effectively all lakes with a mean annual residence time greater than 7 days evade all inflowing  $CO_2$  (Figure 4a). However, evasion efficiency does not track with lake size in the same way: extremely small lakes (in Figure 4b, these are less than 0.004 km<sup>2</sup>) feature a full range of  $CO_2$  evasion efficiencies, but, larger lakes are almost exclusively efficient evaders (Figure 4b). This suggests that there are very small lakes that still have high HRTs (due to low flow rather than a large size) and thus yield near 100% evasion efficiency.

Because most lakes are highly efficient evaders (and functionally all larger lakes evade 100% of CO<sub>2</sub>), there are broad implications for carbon (C) cycling through river networks. We speculate that these larger lakes effectively reset the C cycling through the system, and so any field observations of large amounts of  $CO_2$  downstream of a lake are likely due to internal decomposition of organic matter or a flood event flushing the floodplain rather than from upstream  $CO_2$ . This directly influences our current understandings of differential C inputs along the stream-river continuum, adding additional complexity beyond the simple relationships previously identified between evasion and discharge (Hotchkiss et al., 2015) and residence time (Catalán et al., 2016).

These results agree with previous work highlighting the outsized role of small lentic waterbodies (lakes). Holgerson and Raymond (2016) found extremely small ponds (<0.001 km<sup>2</sup>) are responsible for 15% of CO<sub>2</sub> emissions, while Cheng and Basu (2017) found half of total nitrogen removal from lentic waterbodies occurred in wetlands < 0.001 km<sup>2</sup>. Finally, small connected ponds (<0.01 km<sup>2</sup>) are responsible for 34%, 69%, and 12% of nitrogen, phosphorous, and sediment retention in the northeastern US, respectively, and this is most pronounced in the headwaters (Schmadel et al., 2019). Here, however, we show that at mean annual flow lake influence for potential evasion of terrestrial  $CO_2$  in the Connecticut River system is most pronounced downstream of the headwaters. Finally, we acknowledge that hydrologic connectivity is much broader than the connectivity between rivers and lakes. Upland ponds, the hyporheic zone, groundwater, wetlands, and floodplain zones are all hydrologically connected to the network (Godsey & Kirchner, 2014; Harvey & Gooseff, 2015; Harvey et al., 2019; Schmadel et al., 2019), and future work should focus on coupling these other components with our transport model.

This methodology is repeatable and scalable with a topologically connected network and  $CO_2$  advection/ evasion model in hand. Future workers need only lake surface area extents and mean flow estimates to repeat this analysis in other watersheds or at larger scales. Further, while we only had access to static lake surface areas as defined within the NHD HR, dynamic lake surface area is easily calculated for large lakes, even at global scales, using remote sensing (e.g. Wang et al. 2014). Some novel combination of dynamic lake extent and  $CO_2$  evasion efficiency is likely possible and would add needed nuance to this preliminary work relating lake size and evasion efficiency.

# 4. Conclusions

Lakes are fundamentally connected to river networks and act as stores within the river system for the constituents that move downstream and interact with in-stream processes and the landscape. This study represents a first attempt at testing, on  $CO_2$ , the assertion that river/lake topology influences the form and function of fluvial geochemical processes at network scales (Gardner et al., 2019). Our promising results open the door for related work on lake influences on  $CO_2$  originating from other sources, other GHGs, or even other landscape components. As GHG researchers move toward a more complete understanding of the complex interplay between terrestrial inputs, in-stream processes, evasion to the atmosphere, and landscape geomorphology within river networks, a better understanding of lakes' roles in these processes is necessary.

## Data Availability Statement

The model, results, and code to reproduce our figures are available at https://doi.org/10.5281/ zenodo.4135645. NHD HR hydrography is available at https://www.usgs.gov/core-science-systems/ngp/national-hydrography/access-national-hydrography-products. GRADES is available at http://hydrology.princeton.edu/data/mpan/GRADES/. USGS streamgauge information was obtained using the dataRetrieval R package available at https://github.com/USGS-R/dataRetrieval.



#### Acknowledgments

C. B. Brinkerhoff was supported by a grant to C. J. Gleason through the NSF RoL FELS RAISE program (PI P.A. Raymond, Grant 1840243). We thank the two anonymous reviewers for their thoughtful and helpful comments that greatly improved this manuscript, including suggestions and language for extending our results. We also thank Ed Beighley for developing HRR and Peirong Lin for developing GRADES.

#### References

- Abril, G., Martinez, J.-M., Artigas, L. F., Moreira-Turcq, P., Benedetti, M. F., Vidal, L., et al. (2014). Amazon River carbon dioxide outgassing fuelled by wetlands. *Nature*, 505(7483), 395–398. https://doi.org/10.1038/nature12797
- Aho, K. S., & Raymond, P. A. (2019). Differential response of greenhouse gas evasion to storms in forested and wetland streams. Journal of Geophysical Research: Biogeosciences, 124(3), 649–662. https://doi.org/10.1029/2018JG004750
- Bastviken, D., Tranvik, L. J., Downing, J. A., Crill, P. M., & Enrich-Prast, A. (2011). Freshwater methane emissions offset the continental carbon sink. *Science*, 331(6013), 50. https://doi.org/10.1126/science.1196808
- Battin, T. J., Kaplan, L. A., Findlay, S., Hopkinson, C. S., Marti, E., Packman, A. I., et al. (2008). Biophysical controls on organic carbon fluxes in fluvial networks. *Nature Geoscience*, 1(2), 95–100. https://doi.org/10.1038/ngeo101
- Beighley, R. E., Eggert, K. G., Dunne, T., He, Y., Gummadi, V., & Verdin, K. L. (2009). Simulating hydrologic and hydraulic processes throughout the Amazon River Basin. *Hydrological Processes*, 23(8), 1221–1235. https://doi.org/10.1002/hyp.7252
- Bertuzzo, E., Helton, A. M., Hall, R. O., & Battin, T. J. (2017). Scaling of dissolved organic carbon removal in river networks. Advances in Water Resources, 110, 136–146. https://doi.org/10.1016/j.advwatres.2017.10.009
- Borges, A. V., Darchambeau, F., Teodoru, C. R., Marwick, T. R., Tamooh, F., Geeraert, N., et al. (2015). Globally significant greenhouse-gas emissions from African inland waters. *Nature Geoscience*, 8(8), 637–642. https://doi.org/10.1038/ngeo2486
- Butman, D., & Raymond, P. A. (2011). Significant efflux of carbon dioxide from streams and rivers in the United States. *Nature Geoscience*, *4*(12), 839–842. https://doi.org/10.1038/ngeo1294
- Cael, B. B., Heathcote, A. J., & Seekell, D. A. (2017). The volume and mean depth of Earth's lakes. *Geophysical Research Letters*, 44(1), 209–218. https://doi.org/10.1002/2016GL071378
- Catalán, N., Marcé, R., Kothawala, D. N., & Tranvik, L. J. (2016). Organic carbon decomposition rates controlled by water retention time across inland waters. *Nature Geoscience*, 9(7), 501–504. https://doi.org/10.1038/ngeo2720
- Cheng, F. Y., & Basu, N. B. (2017). Biogeochemical hotspots: Role of small water bodies in landscape nutrient processing. Water Resources Research, 53(6), 5038–5056. https://doi.org/10.1002/2016WR020102
- Cole, J. J., & Caraco, N. F. (2001). Carbon in catchments: Connecting terrestrial carbon losses with aquatic metabolism. *Marine and Freshwater Research*, 52(1), 101–110. https://doi.org/10.1071/mf00084
- Cole, J. J., Prairie, Y. T., Caraco, N. F., McDowell, W. H., Tranvik, L. J., Striegl, R. G., et al. (2007). Plumbing the global carbon cycle: Integrating inland waters into the terrestrial carbon budget. *Ecosystems*, 10(1), 172–185. https://doi.org/10.1007/s10021-006-9013-8
- Crawford, J. T., Lottig, N. R., Stanley, E. H., Walker, J. F., Hanson, P. C., Finlay, J. C., Striegl, R. G. (2014). CO2 and CH4 emissions from streams in a lake-rich landscape: Patterns, controls, and regional significance. *Global Biogeochemical Cycles*, *28*(3), 197–210. https://doi.org/10.1002/2013GB004661
- Czuba, J. A., & Foufoula-Georgiou, E. (2015). Dynamic connectivity in a fluvial network for identifying hotspots of geomorphic change. *Water Resources Research*, *51*(3), 1401–1421. https://doi.org/10.1002/2014WR016139
- Czuba, J. A., Foufoula-Georgiou, E., Gran, K. B., Belmont, P., & Wilcock, P. R. (2017). Interplay between spatially explicit sediment sourcing, hierarchical river-network structure, and in-channel bed material sediment transport and storage dynamics. *Journal of Geophysical Research: Earth Surface*, 122(5), 1090–1120. https://doi.org/10.1002/2016JF003965
- Deemer, B. R., Harrison, J. A., Li, S., Beaulieu, J. J., DelSontro, T., Barros, N., et al. (2016). Greenhouse gas emissions from reservoir water surfaces: A new global synthesis. *BioScience*, 66(11), 949–964. https://doi.org/10.1093/biosci/biw11
- DelSontro, T., Beaulieu, J. J., & Downing, J. A. (2018). Greenhouse gas emissions from lakes and impoundments: Upscaling in the face of global change. *Limnology and Oceanography Letters*, 3(3), 64–75. https://doi.org/10.1002/lol2.10073
- Duvert, C., Butman, D. E., Marx, A., Ribolzi, O., & Hutley, L. B. (2018). CO<sub>2</sub> evasion along streams driven by groundwater inputs and geomorphic controls. *Nature Geoscience*, 11(11), 813–818. https://doi.org/10.1038/s41561-018-0245-y
- Fergus, C. E., Lapierre, J.-F., Oliver, S. K., Skaff, N. K., Cheruvelil, K. S., Webster, K., et al. (2017). The freshwater landscape: Lake, wetland, and stream abundance and connectivity at macroscales. *Ecosphere*, 8(8), e01911. https://doi.org/10.1002/ecs2.1911
- Gardner, J. R., Pavelsky, T. M., & Doyle, M. W. (2019). The abundance, size, and spacing of lakes and reservoirs connected to river networks. Geophysical Research Letters, 46(5), 2592–2601. https://doi.org/10.1029/2018GL080841
- Godsey, S. E., & Kirchner, J. W. (2014). Dynamic, discontinuous stream networks: Hydrologically driven variations in active drainage density, flowing channels and stream order. *Hydrological Processes*, 28(23), 5791–5803. https://doi.org/10.1002/hyp.10310
- Harvey, J., Gomez-Velez, J., Schmadel, N., Scott, D., Boyer, E., Alexander, R., et al. (2019). How hydrologic connectivity regulates water quality in river corridors. JAWRA Journal of the American Water Resources Association, 55(2), 369–381. https://doi.org/10.1111/1752-1688.12691
- Harvey, J., & Gooseff, M. (2015). River corridor science: Hydrologic exchange and ecological consequences from bedforms to basins. Water Resources Research, 51(9), 6893–6922. https://doi.org/10.1002/2015WR017617
- Holgerson, M. A., & Raymond, P. A. (2016). Large contribution to inland water CO2 and CH4 emissions from very small ponds. Nature Geoscience, 9(3), 222–226. https://doi.org/10.1038/ngeo2654
- Holgerson, M. A., Zappa, C. J., & Raymond, P. A. (2016). Substantial overnight reaeration by convective cooling discovered in pond ecosystems. *Geophysical Research Letters*, 43(15), 8044–8051. https://doi.org/10.1002/2016gl070206

Hollister, J., & Milstead, W. B. (2010). Using GIS to estimate lake volume from limited data. Lake and Reservoir Management, 26(3), 194–199. https://doi.org/10.1080/07438141.2010.504321

Hope, D., Palmer, S. M., Billett, M. F., & Dawson, J. J. C. (2004). Variations in dissolved CO2 and CH4 in a first-order stream and catchment: An investigation of soil-stream linkages. *Hydrological Processes*, 18(17), 3255–3275. https://doi.org/10.1002/hyp.5657

- Horgby Å., Segatto, P. L., Bertuzzo, E., Lauerwald, R., Lehner, B., Ulseth, A. J., et al. (2019). Unexpected large evasion fluxes of carbon dioxide from turbulent streams draining the world's mountains. *Nature Communications*, 10(1). https://doi.org/10.1038/s41467-019-12905-z Hotchkiss, E. R., Hall Jr, R. O., Sponseller, R. A., Butman, D., Klaminder, J., Laudon, H., et al. (2015). Sources of and processes controlling
- CO2 emissions change with the size of streams and rivers. *Nature Geoscience*, 8(9), 696–699. https://doi.org/10.1038/ngeo2507 Hu, M., Chen, D., & Dahlgren, R. A. (2016). Modeling nitrous oxide emission from rivers: A global assessment. *Global Change Biology*,
- 22(11), 3566–3582. https://doi.org/10.1111/gcb.13351 Johnson, M. S., Lehmann, J., Riha, S. J., Krusche, A. V., Richey, J. E., Ometto, J. P. H. B., & Couto, E. G. (2008). CO2 efflux from Amazonian bedwater streams represents a similicant fate for deep soil respiration. *Geophysical Research Letters*, 35(17)
- Amazonian headwater streams represents a significant fate for deep soil respiration. *Geophysical Research Letters*, 35(17). https://doi.org/10.1029/2008GL034619 Jones, J. B., Stanley, E. H., & Mulholland, P. J. (2003). Long-term decline in carbon dioxide supersaturation in rivers across the contiguous
- Jones, J. B., Stanley, E. H., & Mulholland, P. J. (2003). Long-term decline in carbon dioxide supersaturation in rivers across the contiguous United States. *Geophysical Research Letters*, 30(10). https://doi.org/10.1029/2003GL017056

- Lauerwald, R., Laruelle, G. G., Hartmann, J., Ciais, P., & Regnier, P. A. G. (2015). Spatial patterns in CO2 evasion from the global river network. *Global Biogeochemical Cycles*, 29(5), 534–554. https://doi.org/10.1002/2014GB004941
- Lauerwald, R., Regnier, P., Figueiredo, V., Enrich-Prast, A., Bastviken, D., Lehner, B., et al. (2019). Natural lakes are a minor global source of N2O to the atmosphere. *Global Biogeochemical Cycles*, 33(12), 1564–1581. https://doi.org/10.1029/2019GB006261
- Lin, P., Pan, M., Beck, H. E., Yang, Y., Yamazaki, D., Frasson, R., et al. (2019). Global reconstruction of naturalized river flows at 2.94 million reaches. Water Resources Research, 55(8), 6499–6516. https://doi.org/10.1029/2019WR025287
- Liu, S., & Raymond, P. A. (2018). Hydrologic controls on pCO2 and CO2 efflux in US streams and rivers. *Limnology and Oceanography Letters*, *3*(6), 428–435. https://doi.org/10.1002/lol2.10095
- Maavara, T., Chen, Q., Van Meter, K., Brown, L. E., Zhang, J., Ni, J., & Zarfl, C. (2020). River dam impacts on biogeochemical cycling. *Nature Reviews Earth & Environment*, 1(2), 103–116. https://doi.org/10.1038/s43017-019-0019-0
- Maavara, T., Lauerwald, R., Laruelle, G. G., Akbarzadeh, Z., Bouskill, N. J., Cappellen, P. V., & Regnier, P. (2019). Nitrous oxide emissions from inland waters: Are IPCC estimates too high? *Global Change Biology*, 25(2), 473–488. https://doi.org/10.1111/gcb.14504
- Marx, A., Dusek, J., Jankovec, J., Sanda, M., Vogel, T., Geldern, R. van, et al. (2017). A review of CO2 and associated carbon dynamics in headwater streams: A global perspective. *Reviews of Geophysics*, 55(2), 560–585. https://doi.org/10.1002/2016RG000547
- Öquist, M. G., Wallin, M., Seibert, J., Bishop, K., & Laudon, H. (2009). Dissolved inorganic carbon export across the soil/stream interface and its fate in a boreal headwater stream. *Environmental Science & Technology*, 43(19), 7364–7369. https://doi.org/10.1021/es900416h
- Raymond, P. A., Hartmann, J., Lauerwald, R., Sobek, S., McDonald, C., Hoover, M., et al. (2013). Global carbon dioxide emissions from inland waters. *Nature*, 503(7476), 355–359. https://doi.org/10.1038/nature12760
- Raymond, P. A., Saiers, J. E., & Sobczak, W. V. (2016). Hydrological and biogeochemical controls on watershed dissolved organic matter transport: Pulse-shunt concept. *Ecology*, 97(1), 5–16. https://doi.org/10.1890/14-1684.1
- Raymond, P. A., Zappa, C. J., Butman, D., Bott, T. L., Potter, J., Mulholland, P., et al. (2012). Scaling the gas transfer velocity and hydraulic geometry in streams and small rivers. *Limnology & Oceanography*, 41–53. https://doi.org/10.1215/21573689-1597669@10.1002/ (ISSN)1939-5590.MethaneVI
- Read, J. S., Hamilton, D. P., Desai, A. R., Rose, K. C., MacIntyre, S., Lenters, J. D., et al. (2012). Lake-size dependency of wind shear and convection as controls on gas exchange. *Geophysical Research Letters*, 39(9). https://doi.org/10.1029/2012GL051886
- Schmadel, N. M., Harvey, J. W., Alexander, R. B., Schwarz, G. E., Moore, R. B., Eng, K., et al. (2018). Thresholds of lake and reservoir connectivity in river networks control nitrogen removal. *Nature Communications*, 9(1). https://doi.org/10.1038/s41467-018-05156-x
- Schmadel, N. M., Harvey, J. W., Schwarz, G. E., Alexander, R. B., Gomez-Velez, J. D., Scott, D., & Ator, S. W. (2019). Small ponds in headwater catchments are a dominant influence on regional nutrient and sediment budgets. *Geophysical Research Letters*, 46(16), 9669–9677. https://doi.org/10.1029/2019GL083937
- Soued, C., del Giorgio, P. A, & Maranger, R. (2016). Nitrous oxide sinks and emissions in boreal aquatic networks in Québec. Nature Geoscience, 9(2), 116–120. https://doi.org/10.1038/ngeo2611
- Stets, E. G., Butman, D., McDonald, C. P., Stackpoole, S. M., DeGrandpre, M. D., & Striegl, R. G. (2017). Carbonate buffering and metabolic controls on carbon dioxide in rivers. *Global Biogeochemical Cycles*, 31(4), 663–677. https://doi.org/10.1002/2016GB005578
- Ulseth, A. J., Hall, R. O., Boix Canadell, M., Madinger, H. L., Niayifar, A., & Battin, T. J. (2019). Distinct air-water gas exchange regimes in low- and high-energy streams. *Nature Geoscience*, 12(4), 259–263. https://doi.org/10.1038/s41561-019-0324-8
- Wang, J., Sheng, Y., & Tong, T. S. D. (2014). Monitoring decadal lake dynamics across the Yangtze Basin downstream of Three Gorges Dam. Remote Sensing of Environment, 152, 251–269. https://doi.org/10.1016/j.rse.2014.06.004
- Wetzel, R. G. (2001). Limnology: Lake and river ecosystems. Gulf Professional Publishing.
- Winterdahl, M., Wallin, M. B., Karlsen, R. H., Laudon, H., Öquist, M., & Lyon, S. W. (2016). Decoupling of carbon dioxide and dissolved organic carbon in boreal headwater streams. *Journal of Geophysical Research: Biogeosciences*, 121(10), 2630–2651. https://doi. org/10.1002/2016JG003420
- Wollheim, W. M., Vörösmarty, C. J., Bouwman, A. F., Green, P., Harrison, J., Linder, E., et al. (2008). Global N removal by freshwater aquatic systems using a spatially distributed, within-basin approach. *Global Biogeochemical Cycles*, 22(2). https://doi.org/10.1029/2007GB002963
- Wood, A. W., Leung, L. R., Sridhar, V., & Lettenmaier, D. P. (2004). Hydrologic implications of dynamical and statistical approaches to downscaling climate model outputs. *Climatic Change*, 62(1–3), 189–216. https://doi.org/10.1023/B:CLIM.0000013685.99609.9e
- Zappa, C. J., McGillis, W. R., Raymond, P. A., Edson, J. B., Hintsa, E. J., Zemmelink, H. J., et al. (2007). Environmental turbulent mixing controls on air-water gas exchange in marine and aquatic systems. *Geophysical Research Letters*, 34(10). https://doi. org/10.1029/2006GL028790
- Zarnetske, J. P., Bouda, M., Abbott, B. W., Saiers, J., & Raymond, P. A. (2018). Generality of hydrologic transport limitation of watershed organic carbon flux across ecoregions of the United States. *Geophysical Research Letters*, 45(21), 11,702–11,711. https://doi. org/10.1029/2018GL080005

## **References From the Supporting Information**

Dingman, S. L. (1984). Fluvial hydrology. New York: W.H. Freeman & Co. Retrieved from https://www.osti.gov/biblio/7061580

- Dingman, L. S. (2007). Analytical derivation of at-a-station hydraulic-geometry relations. *Journal of Hydrology*, 334(1), 17–27. https://doi. org/10.1016/j.jhydrol.2006.09.021
- Messager, M. L., Lehner, B., Grill, G., Nedeva, I., & Schmitt, O. (2016). Estimating the volume and age of water stored in global lakes using a geostatistical approach. *Nature Communications*, 7(1). https://doi.org/10.1038/ncomms13603
- Wanninkhof, R. (1992). Relationship between wind speed and gas exchange over the ocean. *Journal of Geophysical Research: Oceans*, 97(C5), 7373–7382. https://doi.org/10.1029/92JC00188
- Yamazaki, D., Ikeshima, D., Sosa, J., Bates, P. D., Allen, G. H., & Pavelsky, T. M. (2019). MERIT hydro: A high-resolution global hydrography map based on latest topography dataset. NWater Resources Research, 55(6), 5053–5073. https://doi.org/10.1029/2019WR024873